Further Evidence On The Relationship Between Firm Investment And Financial Status

Robert S. Chirinko

(Emory University and CESifo)

Ulf von Kalckreuth

(Deutsche Bundesbank)

Discussion paper 28/02 Economic Research Centre of the Deutsche Bundesbank

November 2002

The discussion papers published in this series represent the authors' personal opinions and do not necessarily reflect the views of the Deutsche Bundesbank.

Deutsche Bundesbank, Wilhelm-Epstein-Strasse 14, 60431 Frankfurt am Main, Postfach 10 06 02, 60006 Frankfurt am Main

Tel +49 69 95 66-1 Telex within Germany 4 1 227, telex from abroad 4 14 431, fax +49 69 5 60 10 71

Please address all orders in writing to: Deutsche Bundesbank, Press and Public Relations Division, at the above address or via fax No. +49 69 95 66-30 77

Reproduction permitted only if source is stated.

ISBN 3-935821-37-9

Summary

The interpretation of the significant relation between business investment spending and cash flow has been controversial. A large body of research has found that investment/cash flow sensitivities are higher for financially constrained firms. This fundamental result underlying the finance constraints hypothesis has been challenged recently by Kaplan, Zingales, and Cleary. The latter author introduces an important new element to this debate by using discriminant analysis, which allows creditworthy firms to be identified using an objective ex-ante criterion based on dividend payouts. Consistent with the Kaplan and Zingales critique, investment/cash flow sensitivities are lower for financially constrained firms. This short paper documents that the use of discriminant analysis does not necessarily lead to lower sensitivities. Our contrasting results are traceable to the use of the firm's creditworthiness as the discriminating variable and appropriate adjustments for endogenous regressors and serially correlated residuals. We document that the investment/cash flow sensitivity is higher for financially constrained firms.

JEL Codes: G32, E22

Zusammenfassung

Die Interpretation der signifikanten Beziehung zwischen unternehmerischen Investitionsausgaben und dem Cash-Flow ist umstritten. Eine größere Anzahl von Forschungsarbeiten kommt zu dem Ergebnis, dass die Sensitivität der Investitionen bezüglich des Cash-Flow bei finanziell beschränkten Unternehmen höher liegt. Dieses für die Theorie finanzieller Beschränkungen grundlegende Resultat wurde in jüngerer Zeit von Kaplan, Zingales und Cleary in Zweifel gezogen. Der letztgenannte Autor führte ein wichtiges neues Element in die Debatte ein: Finanziell beschränkte Unternehmen werden von ihm mit Hilfe eines diskriminanzanalytischen Verfahrens identifiziert, also eines objektiven ex-ante Kriteriums. Im Einklang mit der Kritik von Kaplan und Zingales findet er bei finanziell beschränkten Unternehmen eine geringere Cash-Flow-Sensitivität. Dieses kurze Papier dokumentiert, dass eine diskriminanzanalytische Vorgehensweise nicht notwendigerweise zu einem derartigen Resultat führt. Unsere abweichenden Ergebnisse basieren einerseits auf der Identifikation finanziell beschränkter Unternehmen auf der Basis ihrer Kreditwürdigkeit, andererseits aber auf der Berücksichtigung endogener Regressoren und seriell korrelierter Residuen bei der Schätzung. Wir zeigen auf, dass die Cash-Flow-Sensitivität der Investitionsausgaben bei finanziell beschränkten Unternehmen höher ist.

JEL Klassifikation: G32, E22

Table of Contents

Inti	roduction	1
I.	Estimating Equation	4
Π.	Datasets	6
	A. Creditworthiness Ratio	6
	B. User Costs	7
	C. Financial Statements (UBS)	8
	D. Summary Statistics	8
III.	OLS Estimates	9
IV.	. GMM Estimates	10
	A. Distributed Lag (DL) Specification	10
	B. Autoregressive Distributed Lag (ADL) Specification	11
V.	Summary And Conclusions	12
Ref	ferences	13
Ap	ppendix A: Derivations Of The Distributed Lag (DL) And Autoregressive Distributed Lags (ADL) Estimating Equation	21
Ap	pendix B: The Construction Of User Costs Of Capital For Germany	25
Ap	pendix C: Complete Tables of the Estimated Coefficients	28

Tables

Table I: Summary Statistics For The Full And Split Sample	17
Table II: OLS Parameter Estimates Of Equation (4) (4)	18
Table III: GMM Parameter Estimates Of Equation (4)	19
Table IV: GMM Parameter Estimates of Equation (4)Augmented With Lags Of The Dependent Variable	20
Table C-II: OLS Parameter Estimates Of Equation (4) - Complete Estimates	28
Table C-III: GMM Parameter Estimates Of Equation (4) - Complete Estimates	30
Table C-IV: GMM Parameter Estimates of Equation (4) Augmented With Lags Of The Dependent Variable - Complete Estimates	32

Further Evidence On The Relationship Between Firm Investment And Financial Status *)

Introduction

The impact of finance on firm's investment decisions has been a hotly contested research issue for decades. Since the earliest econometric studies by Tinbergen (1939), Klein (1951), and Meyer and Kuh (1957), investment equations have frequently contained liquidity variables, and these regressors have usually been statistically significant. Despite this and other empirical evidence, the framework of Modigliani and Miller (1958) that was prominent through the mid-1980's questioned the role and interpretation of such financial variables. However, over the past 15 years, work on asymmetric information, costly monitoring, and transactions costs have provided a theoretical basis for the role of finance constraints.¹

A flurry of empirical work has followed on these theoretical developments. In an important paper, Fazzari, Hubbard, and Petersen (FHP, 1988) assess the finance constraints hypothesis in terms of an investment equation that relates investment opportunities ($X_{i,t}$) and an indicator of finance constraints (LIQ_{i,t}) in the following equation,²

$$I_{i,t}/K_{i,t-1} = F[X_{i,t}] + G[LIQ_{i,t}] + u_{i,t},$$
(1)

where $I_{i,t}/K_{i,t-1}$ is the investment/capital ratio for firm i at time t, F[.] and G[.] are functions increasing in their arguments, and $u_{i,t}$ is a stochastic error term. FHP measure $X_{i,t}$ by the Brainard-Tobin's Q ($Q_{i,t}$) and LIQ_{i,t} by the ratio of cash flow to the capital stock ($CF_{i,t}/K_{i,t-1}$), and estimate the following model,

^{*} We gratefully acknowledge the comments of Heinz Herrmann, and the invaluable contribution of Fred Ramb with respect to the construction of our user cost variable. The views expressed in this paper do not necessarily reflect those of the Deutsche Bundesbank or CESifo. All errors, omissions, and conclusions remain the sole responsibility of the authors. Contact: Robert S. Chirinko, Emory University and CESifo, Department of Economics, Emory University, 1602 Mizell Drive, Atlanta, Georgia 30322-2240 USA, PH: 1-404-727-6645, FX: 1-404-727-4639, EM: rchirin@emory.edu; Ulf von Kalckreuth: Deutsche Bundesbank, Economic Research Centre, P.O. Box 100602, Wilhelm-Epstein Strasse 14, D-60006 Frankfurt, Germany, PH: 49-69-9566-2217, FX: 49-69-9566-4317, EM: ulf.von-kalckreuth@bundesbank.de.

¹ See the reviews of the finance constraint literature by Gertler (1988), Schiantarelli (1995), and Hubbard (1998).

 $^{^{2}}$ See, for example, FHP (1988, equation 3) and Cleary (1999, equation 1).

$$I_{i,t}/K_{i,t-1} = \alpha_i + \beta Q_{i,t} + \gamma CF_{i,t}/K_{i,t-1} + u_{i,t},$$
(2)

where α_i , β , and γ are estimated coefficients. There is some evidence of serial correlation in the residuals from this specification.³

An important innovation in the FHP study was assessing the finance constraints hypothesis in terms of the *pattern* of estimated coefficients across classes of firms. In their paper, firms are sorted by the dividend payout ratio under the hypothesis that firms paying-out a lower percentage of their equity income as dividends must face higher costs for external funds. Thus, the γ for low dividend payout firms should be statistically different from zero and *higher than* the γ for the high dividend payout and presumably unconstrained firms. This approach, which has spawned a huge literature with data from different countries and different sortings, tests the null hypothesis of a financially frictionless firm by the significance of the γ 's, and uses their pattern across firms to suggest the alternative of finance constraints.

The FHP paper has been recently criticized in a lively intellectual exchange. Kaplan and Zingales (KZ, 1997) argue that the sorting criterion used by FHP was inappropriate, and misclassified firms' ability to obtain financing. They expand the information for classifying firms by using annual and 10-K reports, public news, and management's discussion of liquidity needs. With their classification, KZ rerun equation (2), and find the opposite result from FHP for investment/cash flow sensitivities – the γ for constrained firms is less than the γ for unconstrained firms. Schiantarelli (1995), Hubbard (1998), and FHP (2000) have offered several arguments against the KZ approach. Chief among these are that some of the additional information used by KZ is highly subjective and that the sample size of constrained firms is too small to support accurate inference. KZ (2000) offer a rejoinder.

Cleary (1999) introduces an important new element into this debate by using discriminant analysis. He defines his reference groups as those firms that either increase or decrease dividend payments. Firms that do not change their dividend payment are excluded. He then computes a discriminant function based on several financial ratios, and uses this function to divide his sample into three mutually exhaustive and exclusive classes – financially constrained, partially financially constrained, and not financially constrained. Cleary has a large sample of 9,219 firm/year observations and, coupled with the use of objective ex-ante

³ FHP (1988, Table 6) find that lags of $Q_{i,t}$ and $CF_{i,t}/K_{i,t-1}$ are statistically significant. Based on formal tests, Blundell, Bond, Devereux, and Schiantarelli (1992) conclude that their residuals are serially correlated and that lagged dependent and independent variables are important in their Q model.

measures generated by the discriminant function, his study addresses the above two criticisms of the KZ analysis. He estimates equation (2) by OLS, and his coefficient estimates support the KZ finding that investment/cash flow sensitivities are lower for financially constrained firms.

This short paper documents that the use of discriminant analysis does not necessarily lead to the Cleary/Kaplan/Zingales (CKZ) finding; the initial FHP result concerning higher cash flow sensitivity for financially constrained firms holds when these firms are identified by means of a discriminant analysis. The differing results are traceable to three factors. First, we use the firm's creditworthiness in place of dividend payments as the discriminant variable. There is rather broad agreement that variations in firm creditworthiness and the resulting wedge between internal and external finance are the key elements in finance constraints models.⁴ We have available unique and confidential data from the Deutsche Bundesbank that allows us to generate a precise indicator of those firms that will face a premium on external finance. We believe that our creditworthiness measure allows for more accurate inferences than is possible with the indirect measures used previously in the literature.⁵ Second, since the right-side variables in (1) or (2) are endogenous, obtaining consistent coefficients requires the use of an instrumental variable estimator. Third, serially correlated residuals may adversely affect inferences. We show that, relative to the preferred Generalized Method Of Moments (GMM) estimator that adjusts for endogenous regressors and serially correlated residuals, OLS biases the results toward the CKZ conclusion of lower cash flow sensitivities for financially constrained firms.

In considering our contribution to the finance constraints debate, we believe it is important to emphasize that, relative to Cleary, our OLS coefficient estimates are based on a different specification with different data. We use data on German firms rather than US firms because of the availability of a creditworthiness measure that has been tested and developed over several years. The econometric specification follows from equation (1) but, since we wish to work with a very large sample in order to amplify the creditworthiness distinctions, we need to include firms not listed on stock exchanges. Hence, a Q model is not feasible. Nonetheless, despite these differences, we are able to reproduce the pattern and magnitude of γ 's obtained by Cleary with OLS. Our study proceeds as follows. Section I discusses the

⁴ For example, FHP (1988, p. 183), KZ (1997, p., 172-173), Bernanke and Gertler (1990, p. 88-89), and Cleary (1999, Abstract).

⁵ Identifying finance constrained firms by their dividend payout behavior was used in the original FHP study. This classification scheme has been controversial because dividend policy may be associated with factors not directly related to creditworthiness, such as investor taxation (Allen, Bernardo, and Welch, 2000), firm maturity (Grullon, Michaely, and Swaminathan, 2002), or corporate governance (Easterbrook, 1984).

model specification and several econometric issues. We estimate a distributed lag (DL) model that includes current and lagged percentage changes in sales and user cost and the current and lagged cash flow/capital ratio. Section II discusses the unique dataset available for this study, and the discriminant analysis that allows us to sort the data to assess the finance constraints hypothesis. Section III contains our OLS results for the creditworthiness ratio and, like CKZ, we find that investment/cash flow sensitivities are lower for financially constrained firms. However, given the endogeneity of the regressors, these OLS results may be biased, and Section IV presents GMM results for the exact same distributed lag model used in Section III. In this case, the cash flow coefficients for constrained and unconstrained firms are not significantly different. Both the GMM and OLS estimates of this DL model appear to be questionable because serial correlation tests suggest that the econometric specification does not provide a satisfactory representation of the dynamic structure of investment. We thus add lagged dependent variables. In this autoregressive distributed lag (ADL) model, we find a marked improvement in model specification and, as in the initial FHP analysis, that investment/cash flow sensitivities are higher for financially constrained firms. Section V contains conclusions.

I. Estimating Equation

Our econometric equation is similar to the model (equation (1)) used frequently in the literature examining finance constraints, and contains separate terms capturing investment opportunities and financial effects. Investment opportunities are measured by Cleary by the market-to-book ratio, a procedure requiring that all firms are listed on a stock exchange. However, to expand the coverage of firms, our dataset also includes non-listed firms. Investment opportunities are specified using the neoclassical theory of capital accumulation, which excludes financing effects. In this model, the desired capital stock is related to a quantity variable (sales, S) and a price variable (the user cost of capital, UC, discussed in Section II.B). This demand for the stock of capital is translated into a demand for the flow of investment by relating the percentage change in capital (or the investment/capital ratio less depreciation, I/K - δ) to current and lagged percentage changes in S and UC. These lag coefficients represent in an unrestricted manner expectations formation and adjustment frictions. (The model can be further generalized by including lagged dependent variables; this extension will be considered in Section IV.B.) The second element in equation (1) concerns the specification of the liquidity variable. As in many studies, we measure financing effects by the ratio of cash flow to the capital stock (CF/K), and capture fixed firm and time effects with firm-specific (α_i) and time-specific (α_t) intercepts. These considerations give rise to the following distributed lag investment specification (DL(H)) of lag length H:

$$\begin{split} I_{i,t}/K_{i,t-1} &= \alpha_{i} + \alpha_{t} - \sum_{h=0}^{H} \beta_{uc,h} \Delta logUC_{i,t-h} + \sum_{h=0}^{H} \beta_{s,h} \Delta logS_{i,t-h} \\ &+ \sum_{h=0}^{H} \gamma_{h} \left(CF_{i,t-h} / K_{i,t-1-h} \right) + u_{i,t} , \end{split}$$
(3)

where the α 's, β 's, and γ 's are estimated coefficients, $u_{i,t}$ is a stochastic error term, i indexes firms, and t indexes time. The other coefficients do not vary across firms. Equation (3) is derived in Appendix A.⁶

The key test for assessing the importance of finance constraints is whether there exists a *differential effect* of cash flow on investment for constrained firms relative to unconstrained firms. We evaluate this test by estimating jointly a set of coefficients for unconstrained firms and a set of differential coefficients between contrasting classes of firms.⁷ For example, consider the coefficient on the contemporaneous cash flow term. In our estimating equation, we replace γ_0 in (3) by $\gamma_0 + \gamma_{\Delta,0} * \text{DFC}_i$, where the latter term is an indicator variable equaling one for firms classified as financially constrained and zero for the contrasting class of unconstrained firms. The estimating equation used in this study is as follows,

$$\begin{split} I_{i,t}/K_{i,t-1} &= \alpha_{i} + \alpha_{t} - \sum_{h=0}^{H} (\beta_{uc,h} + \beta_{\Delta,uc,h} * DFC_{i}) * \Delta logUC_{i,t-h} \\ &+ \sum_{h=0}^{H} (\beta_{s,h} + \beta_{\Delta,s,h} * DFC_{i}) * \Delta logS_{i,t-h} \\ &+ \sum_{h=0}^{H} (\gamma_{h} + \gamma_{\Delta,h} * DFC_{i}) * (CF_{i,t-h} / K_{i,t-1-h}) + u_{i,t} . \end{split}$$

$$(4)$$

This joint approach with the full sample allows us to evaluate in a relatively straightforward manner the differential effect of cash flow in terms of a t-test on the estimated $\gamma_{\Delta,h}$'s. Furthermore, these tests incorporate the covariances among coefficients for constrained and unconstrained firms (e.g., the covariance between γ_0 and $\gamma_{\Delta,0}$). Since our estimating equa-

 $^{^{6}}$ See Chirinko (1993, Section II) for a survey of the extensive literature related to the development of this model.

⁷ A similar approach has been used by Oliner and Rudebusch (1992) and Chirinko and Schaller (1995).

tion contains both current and lagged values of cash flow, the test is based on the sum of the differential coefficients,

$$\eta = \sum_{h=0}^{H} \gamma_{\Delta,h} \,. \tag{5}$$

If η is negative and statistically significant, we obtain the CKZ result that financially constrained firms are less sensitive to cash flow. By contrast, a value of η that is positive and statistically significant is consistent with the original FHP finding that financially constrained firms are more sensitive to cash flow.

II. Datasets

A. Creditworthiness Ratio (CWR)

A unique element in this study is the set of creditworthiness ratios (CWR's) generated by the Bundesbank in performing its rediscounting and lending operations. Bills of exchange are issued by nonfinancial firms, and were frequently presented to the Bundesbank by credit institutions.⁸ When a bill was presented for discounting, the creditworthiness of the issuing firm, as well as all other firms that have held this bill, needed to be determined. In the case of default, liability for payment of the bill falls on any firm that has held the bill. By law, the Bundesbank could only accept bills backed by three parties known to be creditworthy.

The Bundesbank evaluates firms by undertaking a massive effort at collecting financial statement data (discussed in Section II.C) and computing CWR's (*Gesamtkennzahl*) using discriminant analysis.⁹ The two underlying populations are solvent and insolvent firms, where insolvency is indicated by a legal application for bankruptcy. The sample is constructed by first identifying the relatively scarce insolvent firms, and then adding a solvent firm from the same sector. To enhance the statistical properties of the discriminant function, the sample contains an equal number of solvent and insolvent firms. The following information is used to compute the discriminant function: 1) equity/pension provision ratio

⁸ Since the implementation of the Monetary Union on January 1, 1999, the Bundesbank continues to assess creditworthiness in the course of the Eurosystem monetary policy operations, but it no longer rediscounts trade bills.

 $^{^{9}}$ See Deutsche Bundesbank (1999) for further details about the construction of the CWR's and the credit evaluation process.

(adjusted equity capital and pension provisions as a percentage of total capital employed; 2) return on total capital employed (profit before income taxes and before interest payments as a percentage of total capital employed); 3) return on equity (profit before income taxes as a percentage of adjusted equity income); 4) capital recovery rate (net receipts as a percentage of capital invested); 5) net interest payment ratio (net interest as a percentage of turnover); 6) accounting practice (which affects available valuation methods). The weights assigned to these categories are confidential. These ratios are examined by the Bundesbank's Department of Credit, Foreign Exchange, and Financial Markets for outliers. The original CWR's range between -99.9 and 99.9, and have been transformed for this study to vary between 0 and 1.

The discriminant analysis determines two critical values of the CWR that classifies firms into one of three categories: high degree of creditworthiness (Good), low degree of creditworthiness (Endangered), or indeterminate. The proportion of distressed firms in the data used in the discriminant analysis appears representative, and compares favorably to the percentage of failed firms in the overall economy (Deutsche Bundesbank, 1998; Stoess, 2001).

B. User Cost (UC)

The user cost of capital (UC) measures the price of capital, and is comprised of three components,

$$UC = R * P * T, \tag{6}$$

where R, P, and T represent rental, price, and tax terms, respectively. The rental term contains two components, the opportunity cost of funds measured by the real long-term interest rate ($r = i - \pi$, the nominal discount rate (i) less the expected rate of inflation (π) in the price of investment goods) and the economic rate of depreciation (δ). The P term is the price of investment goods relative to the price of output. The two key taxes are the rate of income taxation (reflecting both federal and Länder rates, as well as the "solidarity surcharge") and the present value of the stream of current and future tax depreciation deductions. The user cost variable used in this study is much more complicated than this discussion reveals, and these important details are discussed in Appendix B.

C. Financial Statements (UBS)

The Bundesbank's financial statement database (Unternehmensbilanzstatistik, UBS) constitutes the largest source of accounting data for nonfinancial firms in Germany.¹⁰ About 70,000 annual accounts were collected per year on a strictly confidential basis by the Bundesbank's branch offices. These data were initially subjected to a computer check for logical errors and missing data. Approximately 15,000 accounts had to be excluded because they were incomplete, represented consolidated accounts, or were for firms in sectors (e.g., agriculture) for which no meaningful results could be generated owing to the small amount of available data. Additional checks and corrections for errors were undertaken in the Statistical Department at the Bundesbank's Central Office in Frankfurt before finalizing the UBS database.

The dataset used in estimation is smaller for several reasons. We use data only for firms located in the manufacturing sector of West Germany to avoid any issues of comparability between the western and eastern sections of the country. Sole proprietorships and private partnerships are excluded because their tax treatment depends on personal characteristics that are very difficult to quantify. State dominated corporations are also excluded. The dataset is further reduced by missing values, data cleaning, variable construction involving lags, and outlier control.¹¹ The data extend from 1988 to 1997.¹² We thus have available for our preferred econometric specification containing three lags (equation (4) with H=3) 44,345 datapoints for 6,408 firms. For 1996, these data represent 42% of the total turnover of the West German manufacturing sector and 61% of the total turnover of incorporated firms in all German manufacturing.

D. Summary Statistics

Table 1 contains summary statistics for the variables that will enter the econometric specification; variable definitions are displayed in the table note. In addition, the variable means for the sub-sample defined by CWR are presented in columns 6-8. A noteworthy feature of

¹⁰ This discussion draws on the Deutsche Bundesbank (1998) and Stoess (2001), which contain more detailed descriptions of the UBS data.

¹¹ We control for outliers by discarding the upper and lower 1% tails of sales growth, cash flow divided by the capital stock, and the CWR, and the upper 2% tail of the investment capital ratio.

¹² The beginning year of 1988 is chosen because the definitions of many important financial statement variables were changed in 1986 by the directive harmonizing financial statements in the European Union. For many firms, the changes were not instituted in 1987, and the amount of data available in the UBS is unacceptably low in that year.

the UBS dataset is the extensive coverage of small firms. Nearly one-half of the observations in the full sample are for firms with 100 employees or fewer.

III. OLS Estimates

This section presents results with the OLS estimation strategy used by Cleary. Firm fixedeffects are removed by first-differencing.¹³ Since the focus of the analysis is on the sensitivity of investment to cash flow, only the cash flow coefficients and standard errors are presented in the tables. (Complete tables of the estimated coefficients are presented in Appendix C.) The first column of Table II contains results for the full sample and, as in virtually all investment equations, cash flows is statistically significant. Our sum of the cash flow coefficients η equals 0.137 with a standard error of 0.013.

The important issue dividing CKZ and FHP is the differential sensitivity of cash flow coefficients. Rather than sorting firms by their predicted dividend payout policy as in Cleary, we form sub-samples with a direct measure of the external finance premium, CWR. Firms are sorted into three categories of creditworthiness – Endangered, Good, and Indeterminate – depending on the state in the year before the investment/capital ratio first enters the regression model as a dependent variable. Our large sample permits us to discard the middle group to in order to sharpen the tests between the Endangered and Good firms.

The CWR split sample results confirm CKZ's findings. Estimates for the Good class of firms and the differential between the Endangered and Good classes are presented in columns 2 and 3 of Table II. The sum of coefficient differences between the Endangered and Good firms is -0.066. This η has a p-value of 0.059. Based on these OLS results, financially constrained firms are less sensitive to cash flow, a result that contradicts the findings of FHP and many other published studies.

Our distributed lag model contains several insignificant coefficients that may affect the reported standard errors and p-values for the estimated coefficient sums. To obtain more precise estimates, we retain only those regressors in columns 2 or 3 with t-statistics whose p-values are less than 0.10. This trimmed specification includes the current and once lagged value of cash flow, the current and twice lagged value of the user cost, and the current and thrice lagged values of sales. These trimmed estimates are presented in columns 4 and 5 of

¹³ In Cleary's estimated models, the results are robust to whether firm fixed-effects are removed by mean-differencing or first-differencing (Cleary, fn. 13). We choose the latter method because only a first-difference transformation is appropriate for the GMM estimates to be presented in Section IV.

Table II. For cash flow, the current and once lagged coefficient differences are each negative and statistically significant. Relative to the results in column 3, the key η statistic rises (in absolute value), and has a lower standard error. The null hypothesis of equality between the sum of cash flow coefficient differences has a p-value of 0.000, again confirming the CKZ finding. Interestingly, our η of -0.082 is very similar to the comparable differential of -0.089 reported in Cleary's Table IV.

IV. GMM Estimates

The OLS estimates reported in Table II are subject to two potential problems concerning the error term. First, there is a possible correlation between the error term and the regressors.¹⁴ This endogeneity problem will lead to inconsistent coefficient estimates, and needs to be addressed with an instrumental variables technique. Second, there is some evidence that the residuals are serially correlated. With the First-Difference estimator, white noise errors imply that the residuals for periods t and t-2 will be uncorrelated. We use the Lagrangian Multiplier statistic (LM(2)) proposed by Arellano and Bond (1991) for assessing second-order residual serial correlation. If the model is correctly specified and serially correlated residuals are absent, the p-value for LM(2) should be large. However, as indicated in the bottom row of Table II, serial correlation is evident, and the specification may be suspect. Moreover, serially correlated residuals will make it difficult to find appropriate instruments to correct the endogeneity problem. This section explores the impacts of endogeneity and serial correlation on the estimated coefficients and the inferences about finance constraints.

A. Distributed Lag (DL) Specification

We begin by addressing the issue of endogeneity, but maintain the distributed lag specification (equation 4). The standard approach to addressing endogeneity is to replace OLS with an instrumental variables technique. We use the Generalized Method of Moments (GMM) technique developed by Hansen (1982) and applied to panel data by Arellano and Bond (1991). Apart from controlling for the distorting effects of endogeneity, the GMM estimates also correct for conditional heteroscedasticity. We choose appropriately lagged values of the right-hand variables as instruments: $CF_{i,t-m(t)}/K_{i,t-1-m(t)}$, $\Delta logUC_{i,t-m(t)}$, and $\Delta logS_{i,t-m(t)}$ for $m_t \ge 2$, where m_t is as large as possible given data availability and increases over the sample. A constant and the time dummies are also in the instrument set.

¹⁴ This problem is addressed by Clearly, FHP, and other authors by using the beginning-of-period value of Q. However, the key cash flow variable is defined over period t, and hence is likely to be endogenous.

GMM estimates are presented in Table III, which has the exact same format as Table II. As with OLS, the full sample results in column 1 indicate that cash flow is statistically significant. However, each GMM cash flow coefficient is lower than its OLS counterpart. This result suggests that the OLS results may be adversely affected by a positive correlation between cash flow and shocks to investment spending, the latter captured in the error term. The important change is that the η 's in columns 3 and 5 are now positive. These results imply that investment spending by Endangered firms is more sensitive to cash flow than Good firms. However, the t-statistics are approximately one, and thus there is no statistical grounds for rejecting the hypothesis of equality between the cash flow coefficients for firms sorted by creditworthiness.

The specification of the DL model estimated by GMM is satisfactory in one dimension, but remains unsatisfactory in another. For GMM estimation, Sargan (1958) and Hansen (1982) propose a statistic (SH) for testing overidentifying restrictions. If these restrictions on the instruments are rejected, the p-value for the SH statistic will be small. For the split-sample results, the p-values are in excess of 0.250, and the instruments appear to be valid. However, as in the OLS specification, the p-values for the LM(2) statistics are very close to zero, thus indicating that serial correlation continues to be a problem.

B. Autoregressive Distributed Lag (ADL) Specification

Serially correlated residuals are not uncommon in investment studies (cf. fn. 3), and may reflect misspecified dynamics. In panel models using lagged endogenous regressors as instruments, serially correlated residuals will lead to inconsistent coefficient estimates. In general, coefficients of lagged variables represent in an unrestricted manner expectations formation and adjustment frictions, and including lags of the dependent variable is recommended by the derivation in Appendix A to capture these dynamic effects. The ADL specification follows from equation (4) with the inclusion of an additional sum of lags of the dependent variable – Σ_h ($\lambda_h + \lambda_{\Delta,h}$ *DFC_i) * (I_{i,t-h} / K_{i,t-1-h}), h=1,...,H. With these additional terms, the η statistic capturing the long-run impact of cash flow on investment equals the formula in equation (5) divided by (1- $\Sigma \lambda_h$); standard errors for this nonlinear combination of coefficients are computed by the delta method. Lastly, the instrument list now includes lags of the dependent variable.

GMM estimates of the ADL specification are presented in Table IV. The inclusion of lags of the dependent variable eliminates the serial correlation problem, as the p-values for the LM(2) statistics rise sharply. The p-values for the SH statistic continue to exceed conven-

tional levels of significance. The ADL specification has important implications for cash flow sensitivities. As indicated in column 3, η is positive with a p-value of 0.061.¹⁵ When we remove the insignificant coefficients, η does not change, but the standard error declines. The resulting p-value falls to 0.032. Thus, the null hypothesis that the cash flow coefficients are equal for financially constrained and unconstrained firms is rejected, and the original FHP result is confirmed.

V. Summary And Conclusions

That financial factors are related to business fixed investment is not in question. In many investment equations for many different countries, cash flow and other variables representing financial structure are statistically and economically important. However, the interpretation of this significant relation remains controversial. Fazzari, Hubbard, and Petersen (1998) ignited recent interest by introducing an innovative test that assessed the finance constraints hypothesis in terms of the pattern of estimated coefficients across classes of firms. The FHP paper has been the subject of recent criticisms by Kaplan and Zingales (1997) that, in turn, has led to a series of counter-criticisms. Cleary (1999) addresses some of the important concerns that have been raised by using discriminant analysis on a large dataset.

This short paper presents evidence that the use of discriminant analysis does not necessarily lead to the CKZ finding that cash flow sensitivities are lower for financially constrained firms. Rather than using dividend payouts, we are able to base our discriminant analysis directly on the firm's creditworthiness because of unique and confidential rating data at our disposal. Furthermore, we account for the distorting effects on inferences due to endogenous regressors and serially correlated residuals. Consistent with FHP's original conclusion, we find that firms that are financially constrained, as indicated by their creditworthiness, have higher investment/cash flow sensitivities.

Our results confirm the benefits of using discriminant analysis or other formal techniques for generating objective ex-ante measures for classifying financially constrained firms. The econometric evidence presented here demonstrates the sensitivity of economic inferences to econometric specifications. Additional empirical work is needed to pin-down the proper conclusions to be drawn from investment/cash flow correlations.

¹⁵ This η statistic is appropriate if, as we believe, misspecified dynamics are the source of serial correlation in the residuals. If the serial correlation is due to serial correlation in the error term, then the original formulation of η in equation (5) would be appropriate.

References

- Allen, Franklin, Bernardo, Antonio E., and Welch, Ivo, "A Theory of Dividends Based on Tax Clienteles," *Journal of Finance* 55 (December 2000), 2499-2536.
- Arellano, Manuel, and Bond, Stephen, "Some Tests of Specifications for Panel Data: Monte Carlo Evidence and an Application to Employment Equations," *The Review of Economic Studies* 58 (April 1991), 277-298.
- Auerbach, Alan J., "Taxation, Corporate Financial Policy and the Cost of Capital," *Journal* of Economic Literature 21 (1983), 905-940.
- Bernanke, Ben, and Gertler, Mark, "Financial Fragility and Economic Performance," *The Quarterly Journal of Economics* 105 (February 1990, 87-114.
- Blundell, Richard, Bond, Stephen, Devereux, Michael, and Schiantarelli, Fabio, "Investment and Tobin's Q: Evidence from Company Panel Data," *Journal of Econometrics* 51 (1992), 233-257.
- Chenells, Lucy, and Griffith, Rachel, *Taxing Profits in a Changing World* (London: The Institute for Fiscal Studies, 1997).
- Chirinko, Robert S., *Econometric Models and Empirical Findings for Business Investment* (New York: Basil Blackwell, 1993). (A monograph in the Salomon Brothers Center Series Financial Markets, Institutions & Instruments.)
- Chirinko, Robert S., and Schaller, Huntley, "Why Does Liquidity Matter in Investment Equations?," *Journal of Money, Credit and Banking* 27 (May 1995), 527-548.
- Cleary, Sean, "The Relationship between Firm Investment and Financial Status," *The Journal of Finance* 54 (April 1999), 673-692.
- Deutsche Bundesbank, "The Methodological Basis of the Deutsche Bundesbank's Corporate Balance Sheet Statistics," *Deutsche Bundesbank Monthly Report* 50 (October 1998), 49-64.
- Deutsche Bundesbank, "The Bundesbank's Method of Assessing the Creditworthiness of Business Enterprises," *Deutsche Bundesbank Monthly Report* 50 (January 1999), 51-63.

- Easterbrook, Frank, "Two Agency-Cost Explanations of Dividends," *American Economic Review* 74 (September 1984), 650-659.
- Fazzari, Steven M., Hubbard, R. Glenn, and Petersen, Bruce C., "Financing Constraints and Corporate Investment," *Brookings Papers on Economic Activity* (1988:1), 141-195.
- Fazzari, Steven M., Hubbard, R. Glenn, and Petersen, Bruce C., "Investment-Cash Flow Sensitivities Are Useful: A Comment on Kaplan and Zingales," *Quarterly Journal of Economics* 115 (May 2000), 695-705.
- Gertler, Mark, "Financial Structure and Aggregate Economic Activity: An Overview," *Journal of Money, Credit and Banking* 20 (August 1988, Part 2), p. 559-588.
- Grullon, Gustavo, Michaely, Roni, and Swaminathan, Bhaskaran, "Are Dividend Changes a Sign of Firm Maturity?," *Journal of Business* 75 (2002).
- Hansen, Lars Peter, "Large Sample Properties of Generalized Method of Moments Estimators," *Econometrica* 50 (July 1982), 1029-1054.
- Harhoff, Dietmar, and Ramb, Fred, "Investment and Taxation: Evidence from Firm-Level Panel Data," in Deutsche Bundesbank (ed.), *Investing Today for the World of Tomorrow: Studies on the Investment Process in Europe* (Berlin: Springer-Verlag, 2001), 47-73.
- Hubbard, R. Glenn, "Capital-Market Imperfections and Investment," *Journal of Economic Literature* 36 (March 1998), 193-225.
- Jorgenson, Dale W., "Rational Distributed Lag Functions," *Econometrica* 34 (January 1966), 135-149.
- Kaplan, Steven N., and Zingales, Luigi, "Do Investment-Cash Flow Sensitivities Provide Useful Measures of Finance Constraints?," *Quarterly Journal of Economics* 112 (February 1997), 169-215.
- Kaplan, Steven N., and Zingales, Luigi, "Investment-Cash Flow Sensitivities Are Not Valid Measures of Financing Constraints," *Quarterly Journal of Economics* 115 (May 2000), 707-712.

- King, Mervyn A., and Fullerton, Don (eds.), *The Taxation of Income from Capital* (Chicago: University of Chicago Press (for the NBER), 1984).
- Klein, Lawrence R., "Studies in Investment Behavior," in *Conference on Business Cycles* (New York: National Bureau of Economic Research, 1951), 233-242.
- Meyer, John R., and Kuh, Edwin, *The Investment Decision: An Empirical Study* (Cambridge: Harvard University Press, 1957).
- Modigliani, Franco, and Miller, Merton H., "The Cost of Capital, Corporation Finance and the Theory of Investment," *American Economic Review* 48 (June 1958), 261-297.
- OECD, Taxing Profits in a Global Economy. Domestic and International Issues (Paris: OECD, 1991).
- Oliner, Stephen D., and Rudebusch, Glenn D., "Sources of the Financing Hierarchy for Business Investment," *The Review of Economics and Statistics* 74 (November 1992), 643-654.
- Ramb, Fred, Steuersystem und Investitionstaetigkeit. Eine Paneloekonometrische Untersuchung fuer Deutschland (Taxation and Investment: A Panel Econometric Investigation for Germany) (Baden-Baden: Nomos Verlag, 2003).
- Sargan, John D., "The Estimation of Economic Relationships Using Instrumental Variables," *Econometrica* 26 (1958), 393-415.
- Schiantarelli, Fabio, "Financial Constraints and Investment: A Critical Review of Methodological Issues and International Evidence," in Joe Peek and Eric S. Rosengren (eds.), *Is Bank Lending Important for the Transmission of Monetary Policy?* (Boston: Federal Reserve Bank of Boston, 1995), 177-214.
- Sinn, Hans-Werner, "Systeme der Kapitaleinkommensbesteuerung: Ein Allokationstheorethischer Vergleich" ("Systems of Capital Income Taxation: A Comparison with Respect to Allocation"), in Dieter Boes, Manfred Rose, and Christian Seidl (eds.), Beitraege zur Neueren Steuertheorie: Referate des Finanztheoretischen Seminars im Kloster Neustift bei Brixen 1983 (Berlin: Springer, 1984), 209-238.

- Sinn, Hans-Werner, *Capital Income Taxation and Resource Allocation* (Amsterdam: North-Holland, 1987). (Published in German as *Kapitaleinkommensbesteuerung* (Tue-bingen: Mohr, 1985).)
- Stoess, Elmar, "Deutsche Bundesbank's Corporate Balance Sheet Statistics and Areas of Application," *Schmollers Jahrbuch: Zeitschrift fuer Wirtschafts- und Sozialwissenschaften (Journal of Applied Social Science Studies)* 121 (Spring 2001), 131-137.
- Tinbergen, Jan, "A Method and Its Application to Investment Activity," in *Statistical Testing of Business Cycle Theories*, Vol. 1 (Geneva: League of Nations, 1939).

Full Sample						Split By Creditworthiness			
(1)		(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Variable	Mean	Std. Dev.	25%	Median	75%	Endan- gered	Indeter- minate	Good	
I _t / K _{t-1}	0.1813	0.220	0.0585	0.1161	0.2157	0.1776	0.1891	0.1806	
ΔlogUC _t	0.0222	0.0717	-0.0178	0.0094	0.0644	0.0217	0.0228	0.0222	
UCt	0.1587	0.0183	0.1457	0.1572	0.1697	0.1584	0.1579	0.1589	
∆logSt	0.0206	0.1597	-0.0654	0.0214	0.1068	0.0111	0.0265	0.0218	
St	173.15	1455.12	9.94	26.13	71.25	51.16	57.98	225.31	
CF _t / K _{t-1}	0.2843	0.4941	0.1091	0.1887	0.3308	0.1563	0.2262	0.3267	
ΔCWR_t	0.0004	0.0427	-0.0210	0.005	0.0220	-0.0005	-0.0001	0.0007	
CWR _t	0.5736	0.0618	0.5355	0.5735	0.6150	0.5147	0.5399	0.5946	

 Table I

 Summary Statistics For The Full And Split Samples

Notes To Table I:

The sample contains 44,345 datapoints for 6,408 firms for 1988-1997. The Endangered, Indeterminate, and Good sub-samples contain 18%, 14%, and 68%, respectively, of the total firms in the sample. I_t/K_{t-1} is the investment/capital ratio; UC_t is the user cost of capital; S_t is real sales in millions of Deutschmarks; CF_t/K_{t-1} is the cash flow/capital ratio; CWR_t is the creditworthiness ratio; Δ is the first-difference operator. See Section II and Appendix B for more details about the variables. The sub-sample results in columns 1-3 are sorted by CWR_t: Endangered [Good] represents those datapoints below [above] the lower [higher] critical value of CWR; Indeterminate represents those datapoints between the two critical values.

Table IIOLS Parameter Estimates Of Equation (4)Dependent Variable: It / Kt-1Split Sub-Samples Defined By The Creditworthiness Ratio

	FULL _	SPLIT SAMPLE				
Variable	SAMPLE	Good	Differential	Good	Differential	
	(1)	(2)	(3)	(4)	(5)	
CF _{i,t} /K _{i,t-1}	0.083	0.100	-0.057	0.099	-0.060	
1,1 1,1 1	(0.006)	(0.007)	(0.014)	(0.007)	(0.014)	
CF _{i,t-1} / K _{i,t-2}	0.036	0.038	-0.012	0.036	-0.021	
	(0.005)	(0.007)	(0.014)	(0.006)	(0.013)	
CF _{i,t-2} / K _{i,t-3}	0.014	0.008	0.012			
, ,	(0.005)	(0.006)	(0.012)			
CF _{i,t-3} / K _{i,t-4}	0.004	0.005	-0.008			
<u> </u>	(0.004)	(0.005)	(0.010)			
η	0.137	0.151	-0.066	0.135	-0.082	
•	(0.013)	(0.015)	(0.035)	(0.010)	(0.022)	
	· · · · ·	· · · · ·	· · · · ·	<u> </u>	, , , , , , , , , , , , , , , , , ,	
\mathbf{R}^2	0.028	28 0.032			0.031	
LM(2) p-value	0.000	0 0.000 0.000		0.000		

Notes To Table II:

See the note to Table I for variable definitions. Firm-specific (α_i) and time-specific (α_t) intercepts are also included in the regression equation. Standard errors are in parentheses. The estimates in column 1 are for the full sample. The estimates in columns 2 and 3 are sorted by CWR_t: Good represents coefficients for observations above the higher critical value of CWR; Differential represents coefficients for the difference between the coefficients for Endangered and Good firms, where Endangered firms are those below the lower critical value of CWR. Indeterminate observations have been excluded. The estimates in columns 4 and 5 include only those regressors in columns 2 or 3 with t-statistics whose p-values are less than 0.10. η measures the long-run impact of cash flow on investment; see equation (5). LM(2) is the p-value for the Lagrange Multiplier statistic testing for second-order autocorrelation. Section IV contains a further discussion of this statistic. A complete set of estimated coefficients is contained in Appendix C.

Table IIIGMM Parameter Estimates Of Equation (4)Dependent Variable: It / Kt-1Split Sub-Samples Defined By The Creditworthiness Ratio

	FULL	SPLIT SAMPLE				
Variable	SAMPLE	Good	Differential	Good	Differential	
	(1)	(2)	(3)	(4)	(5)	
CF _{i,t} /K _{i,t-t}	0.073	0.114	0.014	0.107	0.038	
	(0.037)	(0.041)	(0.050)	(0.026)	(0.039)	
CF _{i,t-1} / K _{i,t-2}	0.019	-0.000	0.025			
, ,	(0.014)	(0.017)	(0.022)			
CF _{i,t-2} / K _{i,t-3}	0.011	0.005	0.012			
, ,	(0.005)	(0.006)	(0.010)			
CF _{i,t-3} /K _{i,t-4}	0.008	0.006	-0.001			
, ,	(0.004)	(0.005)	(0.007)			
η	0.110	0.125	0.050	0.107	0.038	
•	(0.027)	(0.031)	(0.046)	(0.026)	(0.039)	
SH p-value	0.017	0	.252		0.270	
LM(2) p-value	0.000	0.000 0.000		0.000		

Notes To Table III:

See the note to Table I for variable definitions. Firm-specific (α_i) and time-specific (α_t) intercepts are also included in the regression equation. Standard errors are in parentheses. The instruments are $CF_{i,t-m(t)}/K_{i,t-1-m(t)}$, $\Delta logUC_{i,t-m(t)}$, and $\Delta logS_{i,t-m(t)}$ for $m_t > 2$, where m_t is as large as possible given data availability and increases over the sample; a constant and α_{τ} are also in the instrument set. The estimates in column 1 are for the full sample. The estimates in columns 2 and 3 are sorted by CWR_t: Good represents coefficients for observations above the higher critical value of CWR; Differential represents coefficients for the difference between the coefficients for Endangered and Good firms, where Endangered firms are those below the lower critical value of CWR. The estimates in columns 4 and 5 include only those regressors in columns 2 or 3 with t-statistics whose p-values are less than 0.10. η measures the long-run impact of cash flow on investment; see equation (5). SH is the p-value for the Sargan-Hansen statistic testing overidentifying restrictions. LM(2) is the p-value for the Lagrange Multiplier statistic testing for second-order autocorrelation. Sections I and IV contain further discussion of these statistics. A complete set of estimated coefficients is contained in Appendix C.

Table IV

GMM Parameter Estimates Of Equation (4) Augmented With Lags Of The Dependent Variable Dependent Variable: I_t / K_{t-1} Split Sub-Samples Defined By The Creditworthiness Ratio

	FULL		SPLIT SAMPLE_					
Variable	SAMPLE	Good	Differential	Good	Differential			
	(1)	(2)	(3)	(4)	(5)			
CF _{i,t} /K _{i,t-1}	0.070	0.071	0.072	0.075	0.079			
· · ·	(0.034)	(0.034)	(0.045)	(0.023)	(0.036)			
CF _{i,t-1} / K _{i,t-2}	0.013	0.005	0.013					
, ,	(0.014)	(0.016)	(0.020)					
CF _{i,t-2} /K _{i,t-3}	0.005	0.005	0.006					
	(0.005)	(0.006)	(0.010)					
CF _{i,t-3} / K _{i,t-4}	0.005	0.003	-0.010					
	(0.004)	(0.005)	(0.007)					
η	0.108	0.092	0.090	0.086	0.090			
-	(0.029)	(0.029)	(0.048)	(0.026)	(0.042)			
SH p-value	0.075	0	.239		0.326			
LM(2) p-value	0.165	0	.132		0.042			

Notes To Table IV:

See the note to Table I for variable definitions. Firm-specific (α_i) and time-specific (α_t) intercepts are also included in the regression equation. Standard errors are in parentheses. The instruments are $CF_{i,t-m(t)}/K_{i,t-1-m(t)}$, $\Delta logUC_{i,t-m(t)}$, $\Delta logS_{i,t-m(t)}$, and $I_{i,t-m(t)}/K_{i,t-1-m(t)}$ for m_t > 2, where m_t is as large as possible given data availability and increases over the sample; a constant and α_t are also in the instrument set. The estimates in column 1 are for the full sample. The estimates in columns 2 and 3 are sorted by CWR_t: Good represents coefficients for observations above the higher critical value of CWR; Differential represents coefficients for the difference between the coefficients for Endangered and Good firms, where Endangered firms are those below the lower critical value of CWR. Indeterminate observations have been excluded. The estimates in columns 4 and 5 include only those regressors in columns 2 or 3 with t-statistics whose p-values are less than 0.10. η measures the long-run impact of cash flow on investment, and equals the formula in equation (5) divided by $(1-\Sigma \lambda_h)$, where the λ_h 's are the coefficients on the lags of the dependent variable; standard errors are computed by the delta method. SH is the p-value for the Sargan-Hansen statistic testing overidentifying restrictions. LM(2) is the p-value for the Lagrange Multiplier statistic testing for second-order autocorrelation. Sections I and IV contain further discussion of these statistics. A complete set of estimated coefficients is contained in Appendix C.

Appendix A: Derivations Of The Distributed Lag (DL) And Autoregressive Distributed Lag (ADL) Estimating Equations

This appendix contains a derivation of the DL and ADL estimating equations. The DL is a special case of the ADL with the lags of the dependent variable suppressed. We first derive the ADL equation, and then show how the DL follows with appropriate restrictions on the coefficients.

The ADL takes as its starting point the demand for the desired capital stock,

$$k_{t}^{*} = \sigma uc_{t}^{*} + \xi s_{t}^{*},$$
 (A-1)

where k_t^* is the log of the desired (or long-run) stock of capital, s_t^* is the log of desired output measured by sales, uc_t^* is the log of the long-run user cost, ξ and σ are long-run elasticities representing the technology, and the t subscript indexes time. For expositional simplicity, we do not include firm-specific subscripts in this derivation. Equation (A-1) follows from a CES production function containing capital and any number of additional factors of production. Note that σ is the elasticity of substitution between labor and capital, and is a key parameter in determining the strength of the interest rate channel of monetary policy on capital formation.

The challenge facing the applied econometrician is to translate the above demand for a stock of capital into the demand for the flow of investment. We begin this translation by assuming that investment equals the change in the desired capital stock,

$$\Delta k_t = \Delta k^*_t, \tag{A-2}$$

where Δ is the first difference operator and k_t is the log of the capital stock. Equation (A-2) is not a satisfactory investment equation because it assumes that the actual capital stock adjusts instantaneously to changes in the desired capital stock. Furthermore, k*_t is unobservable. To derive a useful econometric specification, we introduce dynamics with three assumptions. First, we assume that the adjustment of the actual capital stock (or the investment-capital ratio less depreciation) to its desired level is distributed over time according to the following distributed lag,

$$\Delta k_t = I_t / K_{t-1} - \delta = a(L) \Delta k^*_t, \tag{A-3}$$

where a(L) is a polynomial in the lag operator representing technological constraints such as delivery lags and other adjustment frictions.

Second, following Jorgenson (1966), we assume that a(L) can be approximated by a rational lag, b(L)/(1-c(L)), and rewrite (A-3) as follows,

$$I_{t}/K_{t-1} - \delta = b(L)/(1-c(L)) \Delta k^{*}_{t}, \qquad (A-4)$$
$$= \alpha + b(L) \Delta k^{*}_{t} + c(L) I_{t-1}/K_{t-2},$$

where $\alpha = \delta c(1)$. The b(L)'s, c(L)'s, and α contain technology parameters.

Third, at time t, one can consider the long-run values defining k_t^* in (A-1) as expected values based on current information. We assume that these expectations are determined by the following univariate autoregressions specified as first differences,

$$\Delta uc^*_t = d_{uc}(L) \Delta uc_t, \tag{A-5a}$$

$$\Delta s^*{}_t = d_s(L) \Delta s_t, \tag{A-5b}$$

where the $d_{uc}(L)$'s and $d_s(L)$'s are expectation parameters whose lag lengths need not be equal. Consequently, (A-5) provides a good reason why the length of the distributed lag for user cost and sales variables in our investment equation may not be equal. A disadvantage of (A-5) is that it uses a narrow information set, a point to which we return below.

Combining (A-1)-(A-5) and appending an error term (e_t) , we obtain the following investment equation,

$$I_{t}/K_{t-1} = \alpha + \sigma b(L) d_{uc}(L) \Delta uc_{t} + \xi b(L) d_{s}(L) \Delta s_{t} + c(L) I_{t-1}/K_{t-2} + e_{t}, \qquad (A-6a)$$

$$I_{t}/K_{t-1} = \alpha + f_{uc}(L) \Delta uc_{t} + f_{s}(L) \Delta s_{t} + f_{I/K}(L) I_{t-1}/K_{t-2} + e_{t},$$
(A-6b)

where $f_{uc}(L) = \sigma b(L) d_{uc}(L)$, $f_s(L) = \xi b(L) d_s(L)$, and $f_{I/K}(L) = c(L)$. The f(L)'s represent estimated coefficients, and are a mixture of technology and expectation parameters.

The long-run impacts of changes in the user cost and sales are assessed with the following transformed set of coefficients,

$$\eta_{uc} = f_{uc}(1) / (1 - f_{I/K}(1)), \tag{A-7a}$$

$$\eta_{s} = f_{s}(1) / (1 - f_{I/K}(1)). \tag{A-7b}$$

These long-run elasticities can have a structural interpretation in terms of the technology parameters – σ and ξ – if we impose the following restrictions,

$$b(1) / (1-c(1)) = 1,$$
 (A-8a)

$$d_{uc}(1) = 1,$$
 (A-8b)

$$d_s(1) = 1.$$
 (A-8c)

Equation (A-8a) implies that all orders for capital goods are ultimately delivered. Equations (A-8b) and (A-8c) imply that expected values ultimately move one-for-one with changes in actual values in the information set. Note that the validity of the η 's as long-run elasticities is not dependent on the validity of these assumptions used to identify the structural parameters. However, if we wish to separate technology and expectation parameters and thus, in principle, conduct policy experiments that adhere to the strictures of the Lucas Critique, such identification is essential. The quantitative importance of the Lucas Critique and hence the need to achieve identification has been questioned.

The information set used to form expectations of Δuc_t^* and Δs_t^* can be expanded to include additional variables (z_t 's), and (A-5) is can be generalized as follows,

$$\Delta uc^*_t = d_{uc,uc}(L) \Delta uc_t + d_{uc,z}(L) z_t, \qquad (A-5a')$$

$$\Delta s^*_t = d_{s,s}(L) \Delta s_t + d_{s,z}(L) z_t. \tag{A-5b'}$$

If the z's are variables already appearing as arguments in the investment equation (i.e., Δuc_t , Δs_t , and I_t/K_{t-1}), then the estimating equation is not altered,

$$\begin{split} I_{t}/K_{t-1} &= \alpha + \{ \sigma \ b(L) \ d_{uc,uc}(L) + \xi \ b(L) \ d_{s,uc}(L) \} \Delta uc_{t} \\ &+ \{ \xi \ b(L) \ d_{s,s}(L) + \sigma \ b(L) \ d_{uc,s}(L) \} \ \Delta s_{t} \\ &+ \{ c(L) + \sigma \ b(L) \ d_{uc,I/K}(L) + \xi \ b(L) \ d_{s,I/K}(L) \} \ I_{t-1}/K_{t-2} \ + \ e_{t}, \end{split}$$

$$I_{t}/K_{t-1} &= \alpha \ + \ \beta_{uc}(L) \ \Delta uc_{t} \ + \ \beta_{s}(L) \ \Delta s_{t} \ + \ \lambda(L) \ I_{t-1}/K_{t-2} \ + \ e_{t}, \qquad (A-6b')$$

where the $\beta(L)$'s and $\lambda(L)$'s are defined by the terms in braces in (A-6a'). In this case with more a more general information set, identification of the σ and ξ technology parameters becomes more difficult.

The estimating equation recognizes the possibility that cash flow may also enter as an argument to capture short-term credit constraints (i.e., entering

(A-4)) and/or as an element of the information set used to form expectations of Δuc_t and Δs_t (i.e., as another z entering (A-5)). In either case, current and lagged values of cash flow (scaled by the lagged capital stock) enters as additional regressors. Defining the cash flow coefficients – representing short-term credit constraints – as $\gamma(L)$, we obtain the following equation that is the basis for all of the estimates presented in this paper,

$$\begin{split} I_{t}/K_{t-1} &= \alpha + \beta_{uc}(L) \ \Delta uc_{t} + \beta_{s}(L) \ \Delta s_{t} + \gamma(L) \ CF_{t}/K_{t-1} \\ &+ \lambda(L) \ I_{t-1}/K_{t-2} + e_{t}. \end{split} \tag{A-7}$$

The long-run impacts of user cost, sales, and cash flow are defined as follow,

$$\eta_{uc} = \beta_{uc}(1) / (1 - \lambda(1)),$$
 (A-8a)

$$\eta_{s} = \beta_{s}(1) / (1 - \lambda(1)),$$
 (A-8b)

$$\eta_{\rm cf} = \gamma(1) / (1 - \lambda(1)),$$
 (A-8c)

where η_{uc} and η_s are elasticities and η_{cf} is a semi-elasticity.

The DL model excludes the effect of lags of the dependent variable on the adjustment of the desired capital stock (A-4) and expectations formation (A-5'). Excluding these effects in the above derivation implies the following coefficient restrictions: C(L) = 0, $d_{uc,I/K} = 0$, and $d_{s,I/K} = 0$. With these restrictions, $\lambda(L) = 0$ in equation (A-7). The only differences between this restricted version of equation (A-7) and equation (4) in the text is the inclusion of time-specific intercepts (α_t) and firm-specific subscripts on α_i and the variables.

Appendix B: The Construction Of User Costs Of Capital For Germany¹⁶

The Jorgensonian user cost of capital (see Auerbach (1983) for a derivation) is given by the following formula,

$$UC = \frac{p'}{p} \frac{(1-A)\left(\rho - \pi' + \delta^e\right)}{(1-\tau)} , \qquad (B-1)$$

where p is the output price level, p^{I} is the price of investment goods, A is the present value of depreciation allowances, ρ is the nominal discount rate, π^{I} is the expected rate of investment goods price inflation, δ^{e} is the economic depreciation rate, and τ is the basic corporate tax rate (the rate of tax paid if no profits are distributed). The user cost formula usually reflects investment tax credits determined as a percentage of the price of a purchased asset. During our sample period, no such credits were granted to German firms.

Our construction of user costs takes into account multiple assets, multiple sources of funds, and individual taxation following the approach developed by King and Fullerton (1984), extended by the OECD (1991) and Chenells and Griffith (1997), and applied to the German data by Harhoff and Ramb (2001) and Ramb (2003).

If we distinguish as sources of finance between debt finance, new share issues, and retained earnings, the respective discount rates are given by

ſ

$$\rho = \begin{cases}
i \cdot (1 - \tau) & \text{for debt} \\
i/\theta & \text{for new shares} \\
i \cdot \frac{1 - m}{1 - z} & \text{for retained earnings}
\end{cases}$$
(B-2)

In this expression, the variable θ measures the degree of discrimination between retentions and distributions. It is the opportunity costs of retained earnings in terms of gross dividends forgone; θ equals the additional dividend shareholders would receive if one unit of postcorporate tax earnings were distributed. Furthermore, i is the nominal interest rate, m is the

¹⁶ The user cost of capital for our sample have been constructed on the basis of the computer routines provided by Fred Ramb, who also allowed us to use his tax and depreciation data. Fred's help was crucial and decisive. As we made several changes, however, we have to bear responsibility for the user costs used in this study.

marginal personal tax rate on capital income, and z is the effective tax rate on accrued capital gains.

Between 1977 and 2000, the system of capital income taxation operating in Germany was a split rate system with full imputation. Shareholders who were residents of the Federal Republic received a tax credit in the amount of the corporation tax on distributed profits paid. Ultimately, the tax on capital income on distributed profits was equal to the marginal tax on capital income. For Germany, therefore, the variable θ assumes the value $1/(1-\tau)$. Furthermore, the effective tax rate on accrued capital gains was zero, as capital gains were not taxed after a holding period of one year or more. In this case, the expression for the discount rate reduces to

$$\rho = \begin{cases}
i \cdot (1 - \tau) & \text{for debt} \\
i \cdot (1 - \tau) & \text{for new shares} \\
i \cdot (1 - m) & \text{for retained earnings}
\end{cases}$$
(B-3)

In the system with full imputation that prevailed in Germany from 1977 to 2000, the two types of outside finance are equivalent (Sinn, 1984 and 1987).

To implement this framework and quantify (B-1), we use sector-specific (indexed by j) output price levels $(p_{j,t})$ and depreciation rate $(\delta_{j,t}^e)$. Depreciation rates are calculated from a perpetual inventory equation for sectoral capital stocks and investment flows; rates for 1995-1997 are imputed. The price of capital goods (p_t^I) is an economy-wide deflator dated at the beginning of the year, and the expected inflation rate (π_t^I) measures the rate of growth of p_t^I between the beginning and the end of year t. A_{a,t} is the present value of depreciation allowances as a firm-specific asset-weighted average for three different types of assets (indexed by a): building, machinery and equipment. In each case, finance-specific discount rates are used. (A_{a,t} is computed with an optimal switch from accelerated to straight-line depreciation methods.) The rate of interest rate (i_t) is the average yield to maturity of domestic listed debt securities. The tax rate on retained earnings is calculated as a compound tax combining three different taxes of profits: the basic corporate tax purposes), and the "solidarity surcharge" (s_t, which is levied on all corporate and personal tax payments),

$$\boldsymbol{\tau}_t = (1 + \boldsymbol{s}_t)\boldsymbol{\tau}_t^r (1 - \boldsymbol{g}_t) + \boldsymbol{g}_t, \tag{B-4}$$

As in King and Fullerton, we treat local taxes as a normal tax on profits, ignoring some of its special features.¹⁷ As a marginal tax rate for the shareholder, we used the highest marginal income tax m_t^{max} , again inflated by the solidarity surcharge,

$$m_t = (1+s_t)m_t^{\max}.$$
(B-5)

To combine the different user costs resulting from the three different sources of finance, we use a flow weights defined for the three sources of finance as follows: debt (with total liabilities including the share of borrowed funds in the reserve subject to future taxation), new shares (the first difference of the stock of subscribed capital augmented by share premium or paid-in surplus), and retained earnings (retained earnings with the earned surplus including the share of own funds in the reserves subject to future taxation). For increases of debt, new shares, or retained earnings, the corresponding weight is calculated as a ratio to the sum of *positive* sources of new finance in that year. If a particular weight assumes a negative value, it is set to zero for that year; in each year, the weights sum to unity. For the first year, the respective stock weights are used.

¹⁷ Interest payments are only partly deductible, and the Gewerbesteuer payments are not credited to the shareholders on distribution. The latter, strictly speaking, destroys the basic equivalence between sources of outside finance. The Gewerbesteuer is raised at the local level. Due to data limitations, however, we have to confine ourselves to the mean Gewerbesteuer rate for the whole sample.

Appendix C: Complete Tables of the Estimated Coefficients

Table C-II

OLS Parameter Estimates of Equation (4) - Complete Estimates Dependent Variable: I_t / K_{t-1} Split Sub-Samples Defined by the Creditworthiness Ratio

	FULL		SPLIT SA	MPLE	
Variable	SAMPLE	Good	Differential	Good	Differential
	(1)	(2)	(3)	(4)	(5)
CF _{i,t} /K _{i,t-1}	0.083	0.100	-0.057	0.099	-0.060
	(0.006)	(0.007)	(0.014)	(0.007)	(0.014)
CF _{i,t-1} /K _{i,t-2}	0.036	0.038	-0.012	0.036	-0.021
	(0.005)	(0.007)	(0.014)	(0.006)	(0.013)
CF _{i,t-2} /K _{i,t-3}	0.014	0.008	0.012		
	(0.005)	(0.006)	(0.012)		
CF _{i,t-3} / K _{i,t-4}	0.004	0.005	-0.008		
	(0.004)	(0.005)	(0.010)		
$\eta_{ m cf}$	0.137	0.151	-0.066	0.135	-0.082
	(0.013)	(0.015)	(0.035)	(0.010)	(0.022)
$\Delta logUC_{i,t}$	-0.093	-0.016	-0.071	0.001	-0.062
	(0.031)	(0.029)	(0.057)	(0.026)	(0.056)
ΔlogUC _{i,t-1}	-0.247	-0.195	0.043	-0.176	0.053
, , , , , , , , , , , , , , , , , , ,	(0.038)	(0.029)	(0.057)	(0.026)	(0.054)
$\Delta logUC_{i,t-2}$	-0.091	-0.085	-0.051	-0.057	-0.013
_ ,	(0.037)	(0.031)	(0.059)	(0.023)	(0.049)
∆logUC _{i,t-3}	0.001	-0.028	-0.053		
_ ,	(0.028)	(0.023)	(0.047)		
η_{uc}	-0.430	-0.324	-0.131	-0.232	-0.023
-	(0.110)	(0.093)	(0.170)	(0.062)	(0.128)
$\Delta logS_{i,t}$	0.080	0.089	-0.005	0.090	-0.002
	(0.009)	(0.012)	(0.023)	(0.012)	(0.023)
∆logS _{i,t-1}	0.063	0.079	-0.021	0.080	-0.015
	(0.011)	(0.014)	(0.028)	(0.014)	(0.028)
$\Delta \log S_{i,t-2}$	0.039	0.048	-0.000	0.053	0.011
0	(0.012)	(0.014)	(0.028)	(0.014)	(0.027)
ΔlogS _{i,t-3}	0.013	0.016	0.039	0.019	0.039
0	(0.010)	(0.012)	(0.023)	(0.011)	(0.023)
η _s	0.195	0.233	0.012	0.243	0.034
19	(0.033)	(0.040)	(0.081)	(0.039)	(0.080)
	()	()	(· · · /
\mathbf{R}^2	0.028	0.032		0.031	
LM(2) p-value	0.000		000		0.000

Notes to Table C-II:

See the note to Table I for variable definitions. Firm-specific (α_i) and time-specific (α_t) intercepts are also included in the regression equation. Standard errors are in parentheses. The estimates in column 1 are for the full sample. The estimates in columns 2 and 3 are sorted by CWR_i: Good represents coefficients for observations above the higher critical value of CWR; Differential represents coefficients for the difference between the coefficients for Endangered and Good firms, where Endangered firms are those below the lower critical value of CWR. Indeterminate observations have been excluded. The estimates in columns 4 and 5 include only those regressors in columns 2 or 3 with t-statistics whose p-values are less than 0.10 (the one exception to this trimming rule is for $\Delta logUC_{i,t}$ which has been included in columns 4 and 5 to preserve the continuity of the distributed lag). η_X measures the long-run impact of X (X={CF/K, $\Delta logUC, \Delta logS$ }) on investment; see equation (5). (Note that the η in Table II is equivalent to the η_{cf} in Table C-II.) LM(2) is the p-value for the Lagrange Multiplier statistic testing for second-order autocorrelation. Section IV contains a further discussion of this statistic.

$Table\ C-III \\ GMM\ Parameter\ Estimates\ of\ Equation\ (4)\ -\ Complete\ Estimates\ \\ Dependent\ Variable:\ I_t\ /\ K_{t-1} \\ Split\ Sub-Samples\ Defined\ by\ the\ Creditworthiness\ Ratio$

	FULL		SPLIT	SAMPLE	<u> </u>
Variable	SAMPLE	Good	Differential	Good	Differential
	(1)	(2)	(3)	(4)	(5)
CF _{i,t} /K _{i,t-1}	0.073	0.114	0.014	0.107	0.038
	(0.037)	(0.041)	(0.050)	(0.026)	(0.039)
CF _{i,t-1} / K _{i,t-2}	0.019	-0.000	0.025		
· ·	(0.014)	(0.017)	(0.022)		
$CF_{i,t-2}/K_{i,t-3}$	0.011	0.005	0.012		
· ·	(0.005)	(0.006)	(0.010)		
CF _{i,t-3} /K _{i,t-4}	0.008	0.006	-0.001		
, ,	(0.004)	(0.005)	(0.007)		
η_{cf}	0.110	0.125	0.050	0.107	0.038
•	(0.027)	(0.031)	(0.046)	(0.026)	(0.039)
ΔlogUC _{i,t}	-0.212	-0.335	0.360	-0.316	0.375
<u> </u>	(0.068)	(0.081)	(0.154)	(0.079)	(0.147)
ΔlogUC _{i,t-1}	-0.182	-0.221	0.132	-0.187	0.143
_ ,	(0.039)	(0.044)	(0.093)	(0.035)	(0.073)
∆logUC _{i,t-2}	-0.034	-0.059	-0.040		
_ ,	(0.033)	(0.037)	(0.081)		
∆logUC _{i,t-3}	0.023	-0.001	-0.004		
0,11	(0.026)	(0.029)	(0.062)		
η_{uc}	-0.405	-0.616	0.448	-0.502	0.518
• • •	(0.123)	(0.141)	(0.300)	(0.099)	(0.196)
			1	1	
$\Delta logS_{i,t}$	0.156	0.233	-0.215	0.239	-0.245
	(0.057)	(0.069)	(0.094)	(0.066)	(0.089)
$\Delta logS_{i,t-1}$	0.113	0.123	-0.065	0.122	-0.068
	(0.014)	(0.018)	(0.029)	(0.018)	(0.029)
$\Delta logS_{i,t-2}$	0.079	0.072	-0.034	0.072	-0.036
	(0.011)	(0.014)	(0.023)	(0.014)	(0.023)
$\Delta logS_{i,t-3}$	0.041	0.035	0.010	0.037	0.010
- , -	(0.009)	(0.011)	(0.019)	(0.011)	(0.019)
η _s	0.388	0.462	-0.303	0.469	-0.339
•~	(0.069)	(0.081)	(0.123)	(0.078)	(0.118)
				-	
SH p-value	0.017		.252		0.270
LM(2) p-value	0.000	0	.000	(0.000

Notes to Table C-III:

See the note to Table I for variable definitions. Firm-specific (α_i) and time-specific (α_t) intercepts are also included in the regression equation. Standard errors are in parentheses. The instruments are $CF_{i,t-m}/K_{i,t-1-m}$, $\Delta logUC_{i,t-m}$, and $\Delta logS_{i,t-m}$ for $m_t > 2$, where m_t is as large as possible given data availability and increases over the sample; a constant and α_t are also in the instrument set. The estimates in column 1 are for the full sample. The estimates in columns 2 and 3 are sorted by CWR_t : Good represents coefficients for observations above the higher critical value of CWR; Differential represents coefficients for the difference between the coefficients for Endangered and Good firms, where Endangered firms are those below the lower critical value of CWR. The estimates in columns 4 and 5 include only those regressors in columns 2 or 3 with t-statistics whose p-values are less than 0.10. η_x measures the long-run impact of X (X={CF/K, $\Delta logUC$, $\Delta logS$ }) on investment; see equation (5). (Note that the η in Table III is equivalent to the η_{cf} in Table C-III). SH is the p-value for the Sargan-Hansen statistic testing overidentifying restrictions. LM(2) is the p-value for the Lagrange Multiplier statistic testing for second-order autocorrelation. Sections I and IV contain further discussion of these statistics.

Table C-IV

GMM Parameter Estimates Of Equation (4) Augmented With Lags Of The Dependent Variable - Complete Estimates Dependent Variable: I_t / K_{t-1} Split Sub-Samples Defined By The Creditworthiness Ratio

	FULL	SPLIT SAMPLE			
Variable	SAMPLE	Good	Differential	Good	Differential
	(1)	(2)	(3)	(4)	(5)
CF _{i,t} /K _{i,t-1}	0.070	0.071	0.072	0.075	0.079
	(0.034)	(0.034)	(0.045)	(0.023)	(0.036)
CF _{i,t-1} / K _{i,t-2}	0.013	0.005	0.013		
	(0.014)	(0.016)	(0.020)		
CF _{i,t-2} / K _{i,t-3}	0.005	0.005	0.006		
	(0.005)	(0.006)	(0.010)		
CF _{i,t-3} / K _{i,t-4}	0.005	0.003	-0.010		
	(0.004)	(0.005)	(0.007)		
η _{cf}	0.108	0.092	0.090	0.086	0.090
-	(0.029)	(0.029)	(0.048)	(0.026)	(0.042)
∆logUC _{i,t}	-0.207	-0.309	0.320	-0.288	0.323
	(0.071)	(0.082)	(0.154)	(0.079)	(0.150)
ΔlogUC _{i.t-1}	-0.163	-0.193	0.089	-0.171	0.089
	(0.038)	(0.043)	(0.090)	(0.036)	(0.073)
ΔlogUC _{i,t-2}	-0.014	-0.050	-0.037		
	(0.034)	(0.036)	(0.081)		
ΔlogUC _{i.t-3}	0.038	0.008	0.001		
0,00	(0.027)	(0.030)	(0.062)		
η_{uc}	-0.401	-0.608	0.418	-0.524	0.470
	(0.144)	(0.157)	(0.324)	(0.115)	(0.226)
$\Delta logS_{i,t}$	0.161	0.220	-0.176	0.209	-0.210
	(0.055)	(0.065)	(0.088)	(0.061)	(0.082)
ΔlogS _{i,t-1}	0.095	0.112	-0.069	0.108	-0.066
8 .,. 1	(0.014)	(0.017)	(0.028)	(0.017)	(0.028)
$\Delta \log S_{i,t-2}$	0.065	0.063	-0.042	0.060	-0.038
8 1,1-2	(0.011)	(0.014)	(0.023)	(0.013)	(0.022)
ΔlogS _{i,t-3}	0.033	0.031	0.000	0.030	-0.003
	(0.010)	(0.011)	(0.019)	(0.011)	(0.018)
η _s	0.409	0.478	-0.321	0.466	-0.363
.12	(0.077)	(0.085)	(0.129)	(0.083)	(0.126)

Table C-IV (continued)GMM Parameter Estimates Of Equation (4)Augmented With Lags Of The Dependent Variable - Complete EstimatesDependent Variable - Complete Estimates

	FULL	SPLIT SAMPLE			
Variable	SAMPLE	Good	Differential	Good	Differential
	(1)	(2)	(3)	(4)	(5)
$I_{i,t-1}/K_{i,t-2}$	0.131	0.114	-0.010	0.125	-0.006
	(0.016)	(0.017)	(0.036)	(0.015)	(0.028)
$I_{i,t-2}/K_{i,t-3}$	-0.002	-0.011	-0.002		
	(0.009)	(0.010)	(0.019)		
$I_{i,t-3}/K_{i,t-4}$	0.005	0.003	0.007		
	(0.007)	(0.008)	(0.014)		
$\Sigma_{\rm h} I_{\rm i,t-j}/K_{\rm i,t-h-1}$	0.135	0.106	-0.006	0.125	-0.006
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	(0.025)	(0.027)	(0.062)	(0.015)	(0.028)
SH p-value	0.075	0.2	239	C).326
LM(2) p-value	0.165	0.132 0.042		0.042	

Notes to Table C-IV:

See the note to Table I for variable definitions. Firm-specific (α_i) and time-specific (α_i) intercepts are also included in the regression equation. Standard errors are in parentheses. The instruments are $CF_{i,t-m}/K_{i,t-1-m}$, $\Delta logUC_{i,t-m}$, $\Delta logS_{i,t-m}$, and $I_{i,t-m}/K_{i,t-1-m}$ for $m_t > 2$, where m_t is as large as possible given data availability and increases over the sample; a constant and α_t are also in the instrument set. The estimates in column 1 are for the full sample. The estimates in columns 2 and 3 are sorted by CWR_t: Good represents coefficients for observations above the higher critical value of CWR; Differential represents coefficients for the difference between the coefficients for Endangered and Good firms, where Endangered firms are those below the lower critical value of CWR. Indeterminate observations have been excluded. The estimates in columns 4 and 5 include only those regressors in columns 2 or 3 with t-statistics whose p-values are less than 0.10. η_X measures the long-run impact of X (X={CF/K, $\Delta \log UC$, $\Delta \log S$ }) on investment, and equals the formula in equation (5) divided by $(1-\Sigma \lambda_h)$, where the λ_h 's are the coefficients on the lags of the dependent variable; standard errors are computed by the delta method. (Note that the η in Table IV is equivalent to the η_{cf} in Table C-IV). SH is the p-value for the Sargan-Hansen statistic testing overidentifying restrictions. LM(2) is the p-value for the Lagrange Multiplier statistic testing for second-order autocorrelation. Sections I and IV contain further discussion of these statistics.

The following papers have been published since 2001:

January	2001	Unemployment, Factor Substitution, and Capital Formation	Leo Kaas Leopold von Thadden
January	2001	Should the Individual Voting Records of Central Banks be Published?	Hans Gersbach Volker Hahn
January	2001	Voting Transparency and Conflicting Interests in Central Bank Councils	Hans Gersbach Volker Hahn
January	2001	Optimal Degrees of Transparency in Monetary Policymaking	Henrik Jensen
January	2001	Are Contemporary Central Banks Transparent about Economic Models and Objectives and What Difference Does it Make?	Alex Cukierman
February	2001	What can we learn about monetary policy transparency from financial market data?	Andrew Clare Roger Courtenay
March	2001	Budgetary Policy and Unemployment Dynamics	Leo Kaas Leopold von Thadden
March	2001	Investment Behaviour of German Equity Fund Managers – An Exploratory Analysis of Survey Data	Torsten Arnswald
April	2001	The information content of survey data on expected price developments for monetary policy	Christina Gerberding
May	2001	Exchange rate pass-through and real exchange rate in EU candidate countries	Zsolt Darvas

July	2001	Interbank lending and monetary policy Transmission: evidence for Germany	Michael Ehrmann Andreas Worms
Septembe	r 2001	Precommitment, Transparency and Montetary Policy	Petra Geraats
Septembe	r 2001	Ein disaggregierter Ansatz zur Berechnung konjunkturbereinigter Budgetsalden für Deutschland: Methoden und Ergebnisse *	Matthias Mohr
Septembe	r 2001	Long-Run Links Among Money, Prices, and Output: World-Wide Evidence	Helmut Herwartz Hans-Eggert Reimers
Novembe	r 2001	Currency Portfolios and Currency Exchange in a Search Economy	Ben Craig Christopher J. Waller
December	2001	The Financial System in the Czech Republic, Hungary and Poland after a Decade of Transition	Thomas Reininger Franz Schardax Martin Summer
December	2001	Monetary policy effects on bank loans in Germany: A panel-econometric analysis	Andreas Worms
December	2001	Financial systems and the role of banks in monetary policy transmission in the euro area	M. Ehrmann, L. Gambacorta J. Martinez-Pages P. Sevestre, A. Worms
December	2001	Monetary Transmission in Germany: New Perspectives on Financial Constraints and Investment Spending	Ulf von Kalckreuth
December	2001	Firm Investment and Monetary Trans- mission in the Euro Area	JB. Chatelain, A. Generale, I. Hernando, U. von Kalckreuth P. Vermeulen

* Available in German only.

January	2002	Rent indices for housing in West Germany 1985 to 1998	Johannes Hoffmann Claudia Kurz
January	2002	Short-Term Capital, Economic Transform- ation, and EU Accession	Claudia M. Buch Lusine Lusinyan
January	2002	Fiscal Foundation of Convergence to European Union in Pre-Accession Transition Countries	László Halpern Judit Neményi
January	2002	Testing for Competition Among German Banks	Hannah S. Hempell
January	2002	The stable long-run CAPM and the cross-section of expected returns	Jeong-Ryeol Kim
February	2002	Pitfalls in the European Enlargement Process – Financial Instability and Real Divergence	Helmut Wagner
February	2002	The Empirical Performance of Option Based Densities of Foreign Exchange	Ben R. Craig Joachim G. Keller
February	2002	Evaluating Density Forecasts with an Application to Stock Market Returns	Gabriela de Raaij Burkhard Raunig
February	2002	Estimating Bilateral Exposures in the German Interbank Market: Is there a Danger of Contagion?	Christian Upper Andreas Worms
February	2002	The long-term sustainability of public finance in Germany – an analysis based on generational accounting	Bernhard Manzke
March	2002	The pass-through from market interest rates to bank lending rates in Germany	Mark A. Weth

April	2002	Dependencies between European stock markets when price changes are unusually large	Sebastian T. Schich
May	2002	Analysing Divisia Aggregates for the Euro Area	Hans-Eggert Reimers
May	2002	Price rigidity, the mark-up and the dynamics of the current account	Giovanni Lombardo
June	2002	An Examination of the Relationship Between Firm Size, Growth, and Liquidity in the Neuer Markt	Julie Ann Elston
June	2002	Monetary Transmission in the New Economy: Accelerated Depreci- ation, Transmission Channels and the Speed of Adjustment	Ulf von Kalckreuth Jürgen Schröder
June	2002	Central Bank Intervention and Exchange Rate Expectations – Evidence from the Daily DM/US-Dollar Exchange Rate	Stefan Reitz
June	2002	Monetary indicators and policy rules in the P-star model	Karl-Heinz Tödter
July	2002	Real currency appreciation in acces- sion countries: Balassa-Samuelson and investment demand	Christoph Fischer
August	2002	The Eurosystem's Standing Facilities in a General Equilibrium Model of the European Interbank Market	Jens Tapking
August	2002	Imperfect Competition, Monetary Policy and Welfare in a Currency Area	Giovanni Lombardo

August	2002	Monetary and fiscal policy rules in a model with capital accumulation and potentially non-superneutral money	Leopold von Thadden
September	2002	Dynamic Q-investment functions for Germany using panel balance sheet data and a new algorithm for the capital stock at replacement values	Andreas Behr Egon Bellgardt
October 20	002	Tail Wags Dog? Time-Varying Informa- tion Shares in the Bund Market	Christian Upper Thomas Werner
October 20	002	Time Variation in the Tail Behaviour of Bund Futures Returns	Thomas Werner Christian Upper
November	2002	Bootstrapping Autoregressions with Conditional Heteroskedasticity of Unknown Form	Sílvia Gonçalves Lutz Kilian
November	2002	Cost-Push Shocks and Monetary Policy in Open Economies	Alan Sutherland
November	2002	Further Evidence On The Relationship Between Firm Investment And Financial Status	Robert S. Chirinko Ulf von Kalckreuth

Visiting researcher at the Deutsche Bundesbank

The Deutsche Bundesbank in Frankfurt is looking for a visiting researcher. Visitors should prepare a research project during their stay at the Bundesbank. Candidates must hold a Ph D and be engaged in the field of either macroeconomics and monetary economics, financial markets or international economics. Proposed research projects should be from these fields. The visiting term will be from 3 to 6 months. Salary is commensurate with experience.

Applicants are requested to send a CV, copies of recent papers, letters of reference and a proposal for a research project to:

Deutsche Bundesbank Personalabteilung Wilhelm-Epstein-Str. 14

D - 60431 Frankfurt GERMANY